

Dear Editors and reviewers:

Re: Manuscript ID: 91197 and Title: Single-cell sequencing technology in diabetic wound healing: new insights into the progenitors-based repair strategies

Thank you for your precious comments and advice. These comments are all valuable and very helpful for revising and improving our paper. We have read through comments carefully and have made corrections. As required by the journal, we have corrected the abbreviations, punctuations and references. Based on the instructions provided in your letter, we uploaded the file of the revised manuscript. We would love to thank you for allowing us to resubmit a revised copy of the manuscript and we highly appreciate your time and consideration.

Sincerely,

Yong-Can Huang

Reviewer #1:

Q1. Yellow portion should come under introduction, Abstract should focus on key findings.

Response: We are grateful for the suggestion. Considering the relevant parts have already appeared in the introduction, we have streamlined the abstract to ensure it succinctly conveys the study's value and implications. And the manuscript has been revised and rewritten by SCRIBENDI, and we hope it can meet the journal's standard.